Author's response to reviews

Title: Ethnic Differences in Ischemic Stroke Subtypes in Young-Onset Stroke: The Stroke Prevention in Young Adults Study

Authors:

Megh M Trivedi (megh.trivedi@som.umaryland.edu)
Kathleen A Ryan (kryan@medicine.umaryland.edu)
John W Cole (jcole@som.umaryland.edu)

Version: 2 Date: 27 January 2015

Author's response to reviews:

We revised the manuscript to include line and page numbers as requested.